Author's response to reviews

Title: Does treating obesity stabilize chronic kidney disease?

Authors:
  Sujata Agnani (sagnan@lsuhsc.edu)
  Vidula T Vachharajani (vvachh@lsuhsc.edu)
  Rohit Gupta (rgupta@lsuhsc.edu)
  Naveen K Atray (natray@lsuhsc.edu)
  Tushar J Vachharajani (tvachh@lsuhsc.edu)

Version: 5 Date: 11 June 2005

Author's response to reviews:

Following changes have made to the final version of the manuscript:

1. The conclusion in the abstract has been changed into a sentence rather than a question.
2. Alexander JW et al has been changed to Alexander et al....
3. The highlighting has been removed
4. The references have been changed to correct medline format.